Author's response to reviews

Title: A descriptive study of variables associated with obtaining nipple aspirate fluid in a cohort of non-lactating women

Authors:

Kimberly A Baltzell (kbaltzell@earthlink.net)
Margaret Wrensch (margaret.wrensch@ucsf.edu)
Jennette D. Sison (jennette.sison@ucsf.edu)

Version: 5 Date: 16 October 2006

Author's response to reviews: see over
October 16, 2006

TO:  BioMed Central Editorial

FROM:  Kimberly Baltzell

RE:  MS: 1538142018104106
A descriptive study of variables associated with obtaining nipple aspirate fluid in a cohort of non-lactating women

Dear Biomed Central Editorial Production Team:

Thank you for your confirmation of my manuscript acceptance. I am submitting a revised version based on your suggestions. Please find my responses to your concerns below.

Tables - The format of your tables will not process correctly. The section 'Continuous variables' is a separate table from the following 'Discrete variables' section. Please either re-number all your tables individually (in ascending numerical order rather than as parts A,B,C etc), or edit your tables so that columns are clearly continuous between the tables.

- The tables have been separated into Table 1,2, and 3 and re-formatted per the request above. Table 2 is now the source of data for continuous variables and Table 3 is the source of data for the discrete variables.

Figure title - The image file should not include the title (e.g. Figure 1... etc.) or figure number. The legend and title should be part of the manuscript only.

- The title has been removed from Figure 1 and is included in the manuscript only.

Thank you for your assistance with the final edits. Please contact me if you require any additional information.

Best regards,

Kimberly Baltzell, R.N., Ph.D.